Author’s response to reviews

Title: Cytokine profiling reveals increased serum inflammatory cytokines in idiopathic choroidal neovascularization

Authors:

Shenchao Guo (shenchao_guo@zju.edu.cn)
Houfa Yin (yinhoufa@zju.edu.cn)
Mingjie Zheng (mingjiezheng@zju.edu.cn)
Yizhen Tang (21518179@zju.edu.cn)
Bing Lu (bing_lu@zju.edu.cn)
Xinyi Chen (11418295@zju.edu.cn)
Qiuli Fu (2313009@zju.edu.cn)
Zhenwei Qin (qinzhenwei@zju.edu.cn)
Danni Lyu (elyon9106@zju.edu.cn)
Qiaomei Tang (11518246@zju.edu.cn)
Lifang Zhang (lifangzhang@zju.edu.cn)
Jian Ma (jian_ma@zju.edu.cn)
Li Zhang (zhangli1@zju.edu.cn)
Xiaoyun Fang (xiaoyunfang@zju.edu.cn)

Version: 1 Date: 16 Feb 2019

Author’s response to reviews:

1. We obtained written consent from the participants before the study and we have added this in the Methods and Ethics approval and consent to participate sections.
2. We have removed track changes or highlighting in our manuscript.